Author's response to reviews

Title: An experience of virtual leadership development for human resource managers

Authors:

Karen E Sherk (ksherk@msh.org)
Fiona Nauseda (fionacrimes@yahoo.com)
Sarah Johnson (sjohnson@msh.org)
Delphine Liston (delphine.liston@gmail.com)

Version: 4 Date: 29 December 2008

Author's response to reviews:

The following sections were added to the manuscript:

Competing interests

The authors declare that they have no competing interests.

Authors' contributions

SJ and KES oversee the implementation of all VLDPs for MSH. FN and DL managed the implementation of the two VLDPs discussed in this article. FN and DL also managed and completed the follow-up with VLDP teams discussed in this article. KES drafted the manuscript and SJ reviewed and edited it. All authors read and approved the final manuscript.